

## Format for ANSWERING REVIEWERS

October 21, 2013

Dear Editor,



Please find enclosed the edited manuscript in Word format (file name: 4727-camarareview.doc).

**Title:** Remote ischemic preconditioning as treatment for non-ischemic gastrointestinal disorders: beyond ischemia-reperfusion injury

**Author:** Carlos Rodrigo Camara-Lemarroy

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 4727

The manuscript has been improved according to the suggestions of reviewers:  
Thank you for your comments

I have made the suggested changes and marked them in yellow.

Major comments:

1- I agree with the comment. We have included the statement that post-ERCP pancreatitis and RE are ideal candidates, since RIP is basically a form of pre-treatment. But we included comments on post-conditioning. (pages 12 and 13, new reference number 96).

2- I included information on one of the common protocols. (page 5 final paragraph).

3- I have made the suggested clarification. (page 11 first line)

4- I have combined all tables into one (now table 1).

Minor comments:

1- I corrected the word "we", except in one place where I include the reader in the prose (page 11).

2- I spelled the abbreviations in full in the headings.

3- I corrected all the typos mentioned and made full spell-check with native English reader. We received an A score and B score in language. We believe we have corrected the minor errors.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,



Dr. Carlos Rodrigo Camara-Lemarro  
Department of Internal Medicine  
Universidad Autonoma de Nuevo Leon  
Hospital Universitario "Jose Eleuterio Gonzalez"  
Monterrey, Nuevo Leon, Mexico